Author’s response to reviews

Title: What are the most important unanswered research questions in trial retention? A James Lind Alliance Priority Setting Partnership – The PRioRiTy II (Prioritising Retention in Randomised Trials) Study

Authors:

Dan Brunsdon (dan.brunsdon@abdn.ac.uk)
Linda Biesty (linda.biesty@nuigalway.ie)
Peter Brocklehurst (P.Brocklehurst@bham.ac.uk)
Valerie Brueton (v Brueton@yahoo.co.uk)
Declan Devane (declan.devane@nuigalway.ie)
Jim Elliott (jim.elliott@nhs.net)
Sandra Galvin (sandra.galvin@nuigalway.ie)
Carrol Gamble (C.Gamble@liverpool.ac.uk)
Heidi Gardner (Heidi.gardner.10@aberdeen.ac.uk)
Patricia Healy (patricia.healy@nuigalway.ie)
Kerenza Hood (HoodK1@cardiff.ac.uk)
Joan Jordan (joan@eupati.eu)
Doris Lanz (d.lanz@qmul.ac.uk)
Beccy Maeso (B.Maeso@soton.ac.uk)
Amanda Roberts (AmandaRoberts@ntlworld.com)
Imogen Skene (Imogen.Skene2@bartshealth.nhs.uk)
Irene Soulsby (isoulsby001@hotmail.co.uk)
Derek Stewart (derek_stewart@me.com)
Shaun Treweek (s.treweek@abdn.ac.uk)
Caroline Whiting (C.Whiting@soton.ac.uk)
Andrew Worrall (acworrall@aol.com)
Katie Gillies (k.gillies@abdn.ac.uk)
David Torgerson (david.torgerson@york.ac.uk)
Sharon Wren (s.d.wren@abdn.ac.uk)

Version: 1 Date: 17 Aug 2019

Author’s response to reviews:
Provided in the uploaded cover letter